Massachusetts Institute of Technology Handout 28 6.857: Network and Computer Security November 17, 1998 Professor Ronald L. Rivest

Readings for Lecture 19

Attached are the following papers:

• Paper: PayWord and MicroMint: Two simple micropayment schemes [3]

• Paper: Electronic Lottery Tickets as Micropayments [2]

• Paper: How to Make a Mint: the Cryptography of Anonymous Electronic Cash [1]

References

[1] Laurie Law, Susan Sabett, and Jerry Solinas. How to make a mint: the cryptography of anonymous electronic cash. National Security Agency, Office of Information Security Research and Technology, Cryptology Division, June 1996.

[2] Ronald L. Rivest. Electronic lottery tickets as micropayments. In Rafael Hirschfeld, editor, Financial Cryptography, pages 307–314, Anguilla, British West Indies, 1997. Springer.

[3] Ronald L. Rivest and Adi Shamir. Payword and micromint: Two simple micropayment schemes. (A preliminary version has appeared in CryptoBytes 3,1 (Spring 1996), pages 7–11. The full version is posted by Rivest on the Web.), 1995.